Author's response to reviews

Title: Evaluation of a pulsed-xenon ultraviolet room disinfection device for impact on contamination levels of methicillin-resistant Staphylococcus aureus.

Authors:

Chetan Jinadatha (Chetan.Jinadatha@va.gov)
Ricardo Quezada (rquezada@sw.org)
Thomas W Huber (Thomas.Huber@va.gov)
Jason B Williams (Jason@antimicrobialtestlabs.com)
John E Zeber (John.Zeber@va.gov)
Laurel A Copeland (Laurel.Copeland@va.gov)

Version: 2 Date: 29 April 2013

Author's response to reviews:

Dear Editor,

We have now updated the competing interest section of the manuscript to reflect the funding support as per editorial office instructions.

Thank you,

Chetan Jinadatha.